\begin{table}%ta.2
\caption{\label{table:photo-gwori}Photometric flux measurements of GW~Ori. The value that is marked with the symbol ``$\downarrow$'' is an upper flux limit.}
%\centerline
 {\small    \begin{tabular}{llr}
\hline\hline
      Wavelength [${\rm \mu m}$] & Flux [Jy] & Ref. \\
\hline
      $0.36   $ & $0.068\pm0.014$ & { 1} \\ 	
      $0.44    $ & $ 0.21\pm0.02$ & { 1} \\ 
      $0.55    $ & $ 0.43\pm0.02$ & { 1} \\
      $0.70$ & $0.61\pm0.03$ & { 1}\\
      $0.90    $ & $ 0.82\pm0.05$ & { 1} \\
      $1.25    $ & $ 1.31     \pm 0.04   $ & { 2} \\
      $1.65    $ & $ 1.47     \pm 0.04   $ & { 2} \\
      $2.20    $ & $ 1.47     \pm 0.04   $ & { 2} \\
      $3.50$ & $2.45\pm0.09$ & { 3} \\
      $4.80     $ & $ 2.7\pm1.2$ & { 4} \\
      $18$ & $17\pm3$ & { 5} \\
      $25     $ & $ 20.5     \pm   1.2   $ & { 6} \\
      $60     $ & $ 31.5     \pm   4.1   $ & { 6} \\ 
      $100     $ & $ 35^{\downarrow}$ & { 6} \\   	
      $350     $ & $ 5.0     \pm  0.6   $ & { 7} \\ 
      $450     $ & $ 3.5    \pm  0.4   $ & { 7} \\  
      $800     $ & $0.9     \pm 0.1   $ & { 7} \\  
      $850     $ & $ 1.0     \pm  0.1   $ & { 7} \\ 
      $1100     $ & $0.29     \pm 0.03   $ & { 7} \\
      $1360     $ & $0.26     \pm 0.06   $ & { 7} \\   
      \hline
    \end{tabular}}
\medskip
References~-- {1}: Calvet et~al. (\cite{calvetII}); {2}:~2~MASS catalogue~(Cutri et~al. \cite{cutri}); {3}:~Rydgren \& Vrba (\cite{rydgrenII}); {4}:~Cohen \& Schwartz (\cite{cohen}); {5}:~Cohen (\cite{cohenIV}); {6}:~IRAS catalogue (\cite{iras}); 
          {7}:~Mathieu et~al. (\cite{mathieuII}).
  \end{table}